Author’s response to reviews

Title: A longitudinal study of fruit juice consumption during preschool years and subsequent diet quality and BMI

Authors:

Li Wan (lwan4@mgh.harvard.edu)
Phani Jakkilinki (phanidj@bu.edu)
Martha Singer (msinger@bu.edu)
M. Loring Bradlee (lbradlee@bu.edu)
Lynn Moore (llmoore@bu.edu)

Version: 2 Date: 30 Mar 2020

Author’s response to reviews:

Editor Comments:

1. We notice an area of high textual overlap in the abstract section with the following published source: https://academic.oup.com/cdn/article/2/11/nzy040/5048988
   If this textual overlap is due to an abstract presentation, please acknowledge this in the "Acknowledgement" section of your manuscript.
   This overlap is due to a published abstract for this paper. We have acknowledged this now in the acknowledgements.

2. Please be sure to add an Acknowledgement section in your manuscript.
   Done.

3. In your Authors’ Contribution section, please provide a statement clearly indicating that all authors read and approved the final version of the manuscript.
   Done